Author's response to reviews

Title: Feasibility of a new method using two-dimensional transesophageal echocardiography for aortic annular sizing in patients undergoing transcatheter aortic valve implantation; a case-control study

Authors:

Mohammad A Sherif (mosherif1978@hotmail.com)
Joerg Herold (joerg_herold@hotmail.com)
Wolfram Voelker (Voelker_w@ukw.de)
Octavian Maniuc (Maniuc_o@ukw.de)
Ruediger C Braun-Dullaeus (r.braun-dullaeus@med.ovgu.de)
Christian Praast (christian.praast@st.ovgu.de)
Georg Ertl (ertl_g@ukw.de)

Version: 5 Date: 10 April 2015

Author's response to reviews: see over
Dear editor,

We would like to submit a revised version of our manuscript entitled "Feasibility of a new method using two-dimensional transesophageal echocardiography for aortic annular sizing in patients undergoing transcatheter aortic valve implantation; a case-control study" for publication in the BMC Cardiovascular Disorders Journal.

1- Competing interests were added in page 10.

2- Authors’ contribution in page 11.

3- Acknowledgement in page 12.

Looking forward to hearing from you.

Regards,

On behalf of the authors,

M. Sherif, MF, FESC